Title: High expression level and nuclear localization of Sam68 are associated with progression and poor prognosis in colorectal cancer

Authors:

Wen-Ting Liao (Liaowt2002@gmail.com)
Jun-Ling Liu (liujl@sysucc.org.cn)
Zheng-gen Wang (wangzghd@yahoo.com.cn)
Ling Shi (sling100@163.com)
Yan-Mei Cui (355341388@qq.com)
Ting-Ting Li (vilition@163.com)
Xiao-Hui Zhao (Xiaohuizhao27@gmail.com)
Xiu-Ting Chen (Xiuting2005@163.com)
Yan-Qing Ding (dyg@fimmu.com)
Li-Bing Song (lb.song1@gmail.com)

Version: 3 Date: 4 January 2013

Author’s response to reviews: see over
Jan 4, 2013,
BMC Gastroenterology
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom
Email: editorial@biomedcentral.com
Phone: +44 (0) 20 3192 2009
Fax: +44 (0)20 3192 2010

Dear editors,

Thank you for your decision letter dated Dec 10, 2012, regarding our manuscript (MS: 2642303377933807), entitled “High expression level and nuclear localization of Sam68 are associated with progression and poor prognosis in colorectal cancer” authored by Wen-Ting Liao et al. We have taken all the points raised by the reviewers and added necessary data as the reviewers suggested. We are now re-submitting a revised manuscript with point-to-point responses to the critiques. I hope that you will find that the current version of our manuscript is suitable for publication in BMC Gastroenterology. If you have any question regarding the re-submitted manuscript, please do not hesitate to contact me.

Sincerely yours,

LB Song, MD, PhD,
State Key Laboratory of Oncology in Southern China,
Department of Experimental Research,
Sun Yat-sen University Cancer Center, Guangzhou 510060, P.R. China.
Phone: 86-20-87343192,
Fax: 86-20-87343171,
E-mail: lb.song1@gmail.com